Author’s response to reviews

Title: Comparison of the abrasive properties of two different systems for interproximal enamel reduction: oscillating versus manual strips

Authors:

Francesca Gazzani (francescagazzani@hotmail.it)
Roberta Lione (robertalione@yahoo.it)
Chiara Pavoni (dott.chiarapavoni@gmail.com)
Gianluca Mampieri (gianluca.mampieri@gmail.com)
Paola Cozza (profpaolacozza@gmail.com)

Version: 4 Date: 15 Oct 2019

Author’s response to reviews:

The following sentence has been added in "Ethics approval and consent to participate" section:

All patients signed an informed consent agreement for orthodontic extraction treatment and to allow their teeth to be used for research purposes. However, ethical approval was not mandatory for the retrospective design of the present in-vitro research (F. Suna Kıraç. Is Ethics Approval Necessary for all Trials? A Clear But Not Certain Process. Mol Imaging Radionucl Ther. 2013 Dec; 22(3): 73–75).